Author's response to reviews

Title: The family as a determinant of stunting in children living in conditions of extreme poverty: a case-control study.

Authors:

Hortensia Reyes (hortensia.reyes@imss.gob.mx)
Ricardo Perez-Cuevas (ricardo.perezcuevas@imss.gob.mx)
Araceli Sandoval (rabasp@yahoo.com)
Raul Castillo (racastillora@hotmail.com)
Jose I Santos (jisantos@himfg.com.mx)
Svetlana Doubova (svetlana.doubova@imss.gob.mx)
Gonzalo Gutierrez (gonzalo.gutierrez@imss.gob.mx)

Version: 4 Date: 18 November 2004

Author's response to reviews:

Dear Editor,
According to your request the formatting changes have been made.